%version 2 - created by Dima on March 21, 2017 to be published in Israel Journal.
%version 1 - created by Dima on Jan 2, 2017 to put on the arXiv.
%\documentclass[10pt,a4paper]{article}%
\documentclass[11pt,a4paper]{amsart}%
%\usepackage{tocloft}
%\ProvidesPackage{MyComm}

%\usepackage{showkeys}
\usepackage{amsfonts,amsmath,amssymb}
\usepackage{amssymb,amsthm,amsxtra}
\usepackage[usenames]{color}
\usepackage{amscd}
\usepackage{amsthm}
\usepackage{amsfonts}
\usepackage{amssymb}
\usepackage{amsmath}
\usepackage{graphicx}
\usepackage{hyperref}
\usepackage{enumerate}%
\usepackage[all]{xy}
%\usepackage[normalem]{ulem}
\usepackage[usenames,dvipsnames]{xcolor}
\usepackage{mathrsfs}
\usepackage[margin=1.45in]{geometry}
%\usepackage[capitalize]{cleveref}
\usepackage{cleveref}
\usepackage{stackrel}



 \newtheorem*{corollary*}{Corollary}
 \newtheorem*{construction*}{Construction}
 \newtheorem*{definition*}{Definition}
 \newtheorem*{notation*}{Notation}
 \newtheorem*{lemma*}{Lemma}
 \newtheorem*{theorem*}{Theorem}
 \newtheorem*{remark*}{Remark}
 \newtheorem*{example*}{Example}
 \newtheorem*{conjecture*}{Conjecture}
 \newtheorem*{condition*}{Condition}
 \newtheorem*{result*}{Result}
 \newtheorem*{property*}{Property}


 \newtheorem*{cor*}{Corollary}
 \newtheorem*{const*}{Construction}
 \newtheorem*{defn*}{Definition}
 \newtheorem*{notn*}{Notation}
 \newtheorem*{lem*}{Lemma}
 \newtheorem*{thm*}{Theorem}
 \newtheorem*{rem*}{Remark}
 \newtheorem*{exm*}{Example}
 \newtheorem*{conj*}{Conjecture}




 \newtheorem{lemma}{Lemma}[section]
 \newtheorem{proposition}[lemma]{Proposition}
 \newtheorem{remark}[lemma]{Remark}
 \newtheorem{example}[lemma]{Example}
 \newtheorem{theorem}[lemma]{Theorem}
 \newtheorem{definition}[lemma]{Definition}
 \newtheorem{notation}[lemma]{Notation}
 \newtheorem{result}{Result}[section]
 \newtheorem{property}[lemma]{Property}
 \newtheorem{corollary}[lemma]{Corollary}
 \newtheorem{construction}[lemma]{Construction}
 \newtheorem{condition}[lemma]{Condition}
 \newtheorem{conjecture}[lemma]{Conjecture}
 \newtheorem{task}[lemma]{Task}
 \newtheorem{prob}[lemma]{Problem}
 \newtheorem{problem}[lemma]{Problem}
%
% %\newtheorem{thm}[lemma]{Theorem}
% \mynewthm{thm}{lemma}{Theorem}
%
 \newtheorem{thm}[lemma]{Theorem}
 \newtheorem{prop}[lemma]{Proposition}
 \newtheorem{lem}[lemma]{Lemma}
 \newtheorem{defn}[lemma]{Definition}
 \newtheorem{notn}[lemma]{Notation}
 \newtheorem{cor}[lemma]{Corollary}
 \newtheorem{exm}[lemma]{Example}
 \newtheorem{exc}[lemma]{Exercise}
 \newtheorem{conj}[lemma]{Conjecture}
 \newtheorem{rem}[lemma]{Remark}
 \newtheorem{conc}[lemma]{Conclusion}

 \newtheorem{introtheorem}{Theorem}
 \renewcommand{\theintrotheorem}{\Alph{introtheorem}}
 \crefname{introtheorem}{theorem}{theorems}
 \Crefname{introtheorem}{Theorem}{Theorems}
  \newtheorem{introthm}[introtheorem]{Theorem}
   \crefname{introthm}{theorem}{theorems}
 \Crefname{introthm}{Theorem}{Theorems}
%\renewcommand{\theintrotheorem}{\Roman{introtheorem}}
 \newtheorem{introcorollary}[introtheorem]{Corollary}
  \crefname{introcorollary}{corollary}{corollaries}
 \Crefname{introcorollary}{Corollary}{Corollaries}

 \newtheorem{introcor}[introtheorem]{Corollary}
   \crefname{introcor}{corollary}{corollaries}
 \Crefname{introcor}{Corollary}{Corollaries}
 \newtheorem{introconjecture}[introtheorem]{Conjecture}
   \crefname{introconjecture}{conjectures}{conjectures}
 \Crefname{introconjecture}{Conjecture}{Conjectures}
 \newtheorem{introconj}[introtheorem]{Conjecture}
    \crefname{introconj}{conjectures}{conjectures}
 \Crefname{introconj}{Conjecture}{Conjectures}

 \newtheorem{introlem}[introtheorem]{Lemma}
     \crefname{introlem}{lemma}{lemmas}
 \Crefname{introlem}{Lemma}{Lemmas}
 \newtheorem{introremark}[introtheorem]{Remark}
 \crefname{introremark}{remark}{remarks}
 \Crefname{introremark}{Remark}{Remarks}
 \newtheorem{introrem}[introtheorem]{Remark}
  \crefname{introrem}{remark}{remarks}
 \Crefname{introrem}{Remark}{Remarks}

 \newtheorem{introprop}[introtheorem]{Proposition}
   \crefname{introprop}{Proposition}{Propositions}
 \Crefname{introprop}{Proposition}{Propositions}

 \newtheorem{introdefn}[introtheorem]{Definition}
   \crefname{introdefn}{definition}{definitions}
 \Crefname{introdefn}{Definition}{Definitions}
 \newtheorem{intronotn}[introtheorem]{Notation}
   \crefname{intronotn}{notation}{notations}
 \Crefname{intronotn}{Notation}{Notations}
 \newtheorem{introtask}[introtheorem]{Task}
   \crefname{introtask}{task}{tasks}
 \Crefname{introtask}{Task}{Tasks}
 \newtheorem{introprob}[introtheorem]{Problem}
  \crefname{introprob}{problem}{problems}
 \Crefname{introprob}{Problem}{Problems}
 \newtheorem{introquestion}[introtheorem]{Question}
   \crefname{introquestion}{question}{questions}
 \Crefname{introquestion}{Question}{Questions}


 \crefname{theorem}{theorem}{theorems}
 \Crefname{theorem}{Theorem}{Theorems}
  \crefname{thm}{theorem}{theorems}
 \Crefname{thm}{Theorem}{Theorems}
%\renewcommand{\theintrotheorem}{\Roman{introtheorem}}

  \crefname{corollary}{Corollary}{Corollaries}
 \Crefname{corollary}{Corollary}{Corollaries}


   \crefname{cor}{Corollary}{Corollaries}
 \Crefname{cor}{Corollary}{Corollaries}

   \crefname{conjecture}{conjectures}{conjectures}
 \Crefname{conjecture}{Conjecture}{Conjectures}

    \crefname{conj}{conjectures}{conjectures}
 \Crefname{conj}{Conjecture}{Conjectures}


     \crefname{lem}{lemma}{lemmas}
 \Crefname{lem}{Lemma}{Lemmas}


      \crefname{lemma}{Lemma}{Lemmas}
 \Crefname{lemma}{Lemma}{Lemmas}

 \crefname{remark}{remark}{remarks}
 \Crefname{remark}{Remark}{Remarks}

  \crefname{rem}{remark}{remarks}
 \Crefname{rem}{Remark}{Remarks}


   \crefname{rem}{remark}{remarks}
 \Crefname{rem}{Remark}{Remarks}

   \crefname{proposition}{Proposition}{Proposition}
 \Crefname{proposition}{Proposition}{Proposition}

    \crefname{prop}{Proposition}{Propositions}
 \Crefname{prop}{Proposition}{Propositions}

   \crefname{defn}{definition}{definitions}
 \Crefname{defn}{Definition}{Definitions}

   \crefname{notn}{notation}{notations}
 \Crefname{notn}{Notation}{Notations}

   \crefname{task}{task}{tasks}
 \Crefname{task}{Task}{Tasks}

  \crefname{prob}{problem}{problems}
 \Crefname{prob}{Problem}{Problems}

   \crefname{question}{question}{questions}
 \Crefname{question}{Question}{Questions}

\newcommand{\al}{\alpha}
\newcommand{\alp}{\alpha}
\newcommand{\bt}{\beta}
\newcommand{\gam}{\gamma}
\newcommand{\del}{\delta}
\newcommand{\zet}{\zeta}
\newcommand{\tet}{\theta}
\newcommand{\iot}{\iota}
\newcommand{\kap}{\kappa}
\newcommand{\sig}{\sigma}
\newcommand{\ups}{\upsilon}
\newcommand{\ome}{\omega}
\newcommand{\eps}{\varepsilon}
\newcommand{\ph}{\varphi}



\newcommand{\Gam}{\Gamma}
\newcommand{\Del}{\Delta}
\newcommand{\Tet}{\Theta}
\newcommand{\Sig}{\Sigma}
\newcommand{\Ups}{\Upsilon}
\newcommand{\Ome}{\Omega}
\newcommand{\Ph}{\Phi}

\newcommand{\stab}{\operatorname{Stab}}
\newcommand{\Ind}{\operatorname{Ind}}
\newcommand{\ind}{\operatorname{ind}}
\newcommand{\Ext}{\operatorname{Ext}}
\newcommand{\id}{\operatorname{Id}}
\newcommand{\Id}{\operatorname{Id}}
\newcommand{\re}{\operatorname{Re}}
\renewcommand{\Im}{\operatorname{Im}}
\newcommand{\Ker}{\operatorname{Ker}}


\newcommand{\Log}{{\operatorname{Log}}}
\newcommand{\Graph}{{\operatorname{Graph}}}
\newcommand{\jet}{{\operatorname{jet}}}
\newcommand{\Tor}{{\operatorname{Tor}}}
\newcommand{\const}{{\operatorname{const}}}
\newcommand{\Arc}{{\operatorname{Arc}}}
\newcommand{\Span}{{\operatorname{Span}}}
\newcommand{\Aut}{{\operatorname{Aut}}}
\newcommand{\rk}{{\operatorname{rk}}}
\newcommand{\End}{\operatorname{End}}

%\newcommand{\Hom}{\operatorname{Hom}}
%\newop{\Hom}{Hom}
%\snewop{Hom}
%\snewop{Mor}

\newcommand{\GL}{\operatorname{GL}}
\newcommand{\SL}{\operatorname{SL}}
\newcommand{\Sp}{\operatorname{Sp}}
\newcommand{\gl}{{\mathfrak{gl}}}
\newcommand{\sll}{{\mathfrak{sl}}}
\newcommand{\spp}{{\mathfrak{sp}}}
\newcommand{\oH}{\operatorname{H}}
\newcommand{\HC}{\operatorname{HC}}
\newcommand{\Mp}{\operatorname{Mp}}
\newcommand{\Sym}{\operatorname{Sym}}
\newcommand{\Spec}{\operatorname{Spec}}
\newcommand{\Lie}{\operatorname{Lie}}
\newcommand{\ad}{\operatorname{ad}}
\newcommand{\Ad}{\operatorname{Ad}}
\newcommand{\sign}{\operatorname{sign}}
\newcommand{\WF}{\operatorname{WF}}

\newcommand{\Supp}{\operatorname{Supp}}
\newcommand{\supp}{\operatorname{Supp}}
\newcommand{\Stab}{\operatorname{Stab}}
\newcommand{\Mat}{\operatorname{Mat}}

%\renewcommand{\tr}{\operatorname{Tr}}
\newcommand{\Tr}{\operatorname{Tr}}
%\newcommand{\det}{\operatorname{Det}}
\newcommand{\Det}{\operatorname{Det}}
\newcommand{\Hom}{\operatorname{Hom}}
\newcommand{\Iso}{\operatorname{Iso}}
\newcommand{\diag}{\operatorname{diag}}
\newcommand{\sgn}{\operatorname{sgn}}
\newcommand{\irr}{\operatorname{irr}}


\newcommand{\bC}{\mathbb{C}}
\newcommand{\bN}{\mathbb{N}}
\newcommand{\bR}{\mathbb{R}}
\newcommand{\bQ}{\mathbb{Q}}
\newcommand{\bZ}{\mathbb{Z}}



\newcommand{\R}{\mathbb{R}}
\newcommand{\CC}{\mathbb{C}}
\newcommand{\RR}{\mathbb{R}}
\newcommand{\ZZ}{\mathbb{Z}}
\newcommand{\OO}{\mathcal{O}}
\newcommand{\NN}{\mathcal{N}}
\newcommand{\distr}{\mathcal{D}}
\newcommand{\gotG}{\mathfrak{g}}
\newcommand{\raw}{\rightarrow}
\newcommand{\xx}{\mathfrak{X}}
%\newcommand{\Sym}{\text{Sym}}


%\snewop{Stab}
%\snewop{supp}
%\snewop{Supp}

%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%other
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%

\newcommand{\ctp}{\widehat{\otimes}}
\newcommand{\Sc}{\cS}
\newcommand{\Omtop}{\Omega^{top}}

\newcommand{\rep}{representation}
\newcommand{\irrep}{irreducible \rep}
\newcommand{\Rep}{Representation}
\newcommand{\Irrep}{Irreducible \rep}
\newcommand{\Arch}{Archimedean}
\newcommand{\Alg}{Algebraic}
\newcommand{\alg}{algebraic}
\newcommand{\Ana}{Analytic}
\newcommand{\ana}{analytic}
\newcommand{\var}{variety}
\newcommand{\Var}{Variety}
\newcommand{\algvar}{\alg\ \var}
\newcommand{\Algvar}{\Alg\ \var}
\newcommand{\anvar}{\ana\ \var}
\newcommand{\Anvar}{\Ana\ \var}
\newcommand{\man}{manifold}
\newcommand{\Man}{Manifold}
\newcommand{\algman}{\alg\ \man}
\newcommand{\Algman}{\Alg\ \man}
\newcommand{\anman}{\ana\ \man}
\newcommand{\Anman}{\Ana\ \man}

\newcommand{\Sch}{Schwartz}
\newcommand{\SchFun}{Schwartz function}
\newcommand{\Four}{Fourier}
\newcommand{\FourTran}{Fourier transform}
\newcommand{\Fre}{Fr\'echet}
\newcommand{\FreSp}{Fr\'echet space}
\newcommand{\tvs}{topological vector space}
\newcommand{\Tvs}{Topological vector space}
\newcommand{\NFS}{Nucler \FreSp}
\newcommand{\Ns}{nuclear space}
\newcommand{\ns}{Nuclear space}
\newcommand{\lcs}{locally convex space}
\newcommand{\Lcs}{Locally convex space}
\newcommand{\et}{\'etale}
\newcommand{\Et}{\'Etale}
\newcommand{\onto}{\twoheadrightarrow}
\newcommand{\into}{\hookrightarrow}

\providecommand{\fb}{\mathfrak{b}}
\providecommand{\fh}{\mathfrak{h}}
\providecommand{\fg}{\mathfrak{g}}
\providecommand{\fm}{\mathfrak{m}}
\providecommand{\fs}{\mathfrak{s}}

\providecommand{\ft}{\mathfrak{t}}
\providecommand{\fz}{\mathfrak{z}}
\providecommand{\fS}{\mathfrak{S}}
\providecommand{\fG}{\mathfrak{G}}

\providecommand{\cA}{\mathcal{A}}
\providecommand{\cB}{\mathcal{B}}
\providecommand{\cC}{\mathcal{C}}
\providecommand{\cD}{\mathcal{D}}
\providecommand{\cE}{\mathcal{E}}
\providecommand{\cF}{\mathcal{F}}
\providecommand{\cG}{\mathcal{G}}
\providecommand{\cH}{\mathcal{H}}
\providecommand{\cI}{\mathcal{I}}
\providecommand{\cJ}{\mathcal{J}}
\providecommand{\cK}{\mathcal{K}}
\providecommand{\cL}{\mathcal{L}}
\providecommand{\cM}{\mathcal{M}}
\providecommand{\cN}{\mathcal{N}}
\providecommand{\cO}{\mathcal{O}}
\providecommand{\cR}{\mathcal{R}}
\providecommand{\cS}{\mathcal{S}}
\providecommand{\cT}{\mathcal{T}}
\providecommand{\cU}{\mathcal{U}}
\providecommand{\cV}{\mathcal{V}}


\providecommand{\g}{\mathfrak{g}}

\newcommand{\simpAr}[2][r]{%
%\ar@{}[#1]|-*[@]_{\csname rot#1\endcsname{#2}}%
\ar@{}[#1]|-*[@]_{#2}%
}

\newcommand{\Dima}[1]{{{#1}}}
\newcommand{\DimaA}[1]{{{#1}}}
\newcommand{\DimaB}[1]{{{#1}}}
%\newcommand{\DimaC}[1]{{\color{blue}{#1}}}
\newcommand{\Rami}[1]{{{#1}}}
\newcommand{\RamiA}[1]{{{#1}}}
\newcommand{\RamiB}[1]{{{#1}}}
\newcommand{\RamiC}[1]{{{#1}}}
\newcommand{\RamiD}[1]{{{#1}}}
\newcommand{\RamiE}[1]{{{#1}}}
\newcommand{\RamiF}[1]{{{#1}}}
\newcommand{\EitanA}[1]{{{#1}}}
%\newcommand{\EitanB}[1]{{\color{green}{#1}}}
%\newcommand{\nextversion}[1]{{\color{magenta}{Next version: #1}}} %?? Do not forget!
%\newcommand{\nextversion}[1]{} %?? Do not forget!
%\newcommand{\nir}[1]{{#1}}
%\newcommand{\RamiQ}[1]{{\color{magenta}{Rami: #1}}}



\newcommand{\proofend}{\hfill$\Box$\smallskip}
\newcommand{\lbl}[1]{\label{#1}}
\newcommand{\ot}{\leftarrow}


\def\smashedlongrightarrow{\setbox0=\hbox{$\longrightarrow$}\ht0=1pt\box0}
\def\risom{\buildrel\sim\over{\smashedlongrightarrow}}

%\newcommand{\st}[1]{\stackrel{#1}{\smashedlongrightarrow}}

\def\smashedst{\setbox0=\hbox{$\rightrightarrows$}\ht0=4pt\box0}
\newcommand{\sst}[1]{\stackrel{#1}{\smashedst}}
%\def\st#1#2{\stackrel[#2]{#1}{\rightrightarrows}}
\def\st#1#2{\stackrel[#2]{#1}{\smashedst}}




\def\twomorphism{\setbox0=\hbox{$\Rightarrow$}\ht0=4pt\box0}
\newcommand{\twomor}[1]{\stackrel{#1}{\twomorphism}}

\oddsidemargin=-0.5cm

\evensidemargin=-0.5cm

\textwidth 17.3cm

\textheight=23cm
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%\newop{\im}{Im}
%\newop{\ker}{Im}
%\newop{\re}{re}
%\snewop{Stab}

%\newcommand{\Opr}[2]{%
%\cmd{#2}{\opr{#1}}%
%}


\begin{document}
\begin{thebibliography}{AAAAAA}



\bibitem[AG08]{AG} A. Aizenbud, D.
Gourevitch, {\it Schwartz functions on Nash Manifolds,}
International Mathematics Research Notices, Vol. 2008, n.5,
Article ID rnm155, 37 pages. DOI: 10.1093/imrn/rnm155.

\bibitem[\href{http://projecteuclid.org/DPubS?service=UI&version=1.0&verb=Display&handle=euclid.dmj/1251120011}{AG09a}]{AG_HC}
A. Aizenbud, D. Gourevitch:{\it Generalized Harish-Chandra descent, Gelfand pairs and an
Archimedean analog of Jacquet-Rallis' Theorem.} arXiv:0812.5063v5[math.RT]. Postprint, originally published in Duke Mathematical Journal, 149/3 (2009).

\bibitem[AG09b]{AG_MOT}
A. Aizenbud,  D. Gourevitch, {\it
Multiplicity one theorem for
$(\mathrm{GL}_{n+1}(\R),\mathrm{GL}_n(\R))$}, Selecta Mathematica,
{\bf 15}, No. 2, pp. 271-294 (2009). See also
arXiv:0808.2729[math.RT].

\bibitem[AG10]{AG_DR} A. Aizenbud, D. Gourevitch:
{\it De-Rham theorem and Shapiro lemma for Schwartz functions on Nash manifolds.}
Israel Journal of Mathematics,
\textbf{171} pp 155-188 (2010). See also
arXiv:0802.3305[math.AG].

\bibitem[AG13]{AG_ST} A. Aizenbud,  D. Gourevitch: {\it
Smooth Transfer of Kloosterman Integrals}, American Journal of Mathematics \textbf{135}, n. 1,  pp. 143-182 (2013). See also arXiv:1001.2490[math.RT].
%
% \bibitem[AG15]{AG_ZR} A. Aizenbud, D. Gourevitch:
% {\it Vanishing of certain equivariant distributions on spherical spaces,}
% Mathematische Zeitschrift, {\bf 279}, Issue 3-4, 745-751 (2015).
% See also
% arXiv:1311.6111.

%\bibitem[AGb]{AG_Zdense} A. Aizenbud, D. Gourevitch:
%{\it Density of $\fz$-eigen distributions on $p$-adic groups.,}
%preprint.
%available at \url{http://www.wisdom.weizmann.ac.il/~dimagur/Publication_list.html}.


% \bibitem[AGc]{AG_OrbitDense} A. Aizenbud, D. Gourevitch:
% {\it Density of regular orbital integrals and sets and sets neglected by distributions,}
% preprint.
%available at \url{http://www.wisdom.weizmann.ac.il/~dimagur/Publication_list.html}.



%\bibitem[AGM16]{AGM}A. Aizenbud, D. Gourevitch, A. Minchenko
%{\it Holonomicity of relative characters and applications to multiplicity bounds for spherical pairs,} Selecta Mathematica {\bf 22}, n. 4, 2325-2345 (2016). DOI:10.1007/s00029-016-0276-4 (2016).  See also arXiv:1501.01479.

\bibitem[AGRS10]{AGRS} A. Aizenbud,  D. Gourevitch, S. Rallis, G. Schifmann : {\it
Multiplicity One Theorems}. Annals of
Mathematics, {\bf 172} (2010), no. 2, 1407-1434. See also arXiv:0709.4215[math.RT].

\bibitem[AGS15]{AGS_Zeig} A. Aizenbud, D. Gourevitch, E.Sayag:
{\it $\mathfrak{z}$-finite distributions on p-adic groups}, Advances in Mathematics
 \textbf{285},  pp. 1376-1414 (2015). See also arXiv:1405.2540.

\bibitem[AHR]{StackLuna} J. Alper, J. Hall, D. Rydh
 {\it A Luna slice theorem for algebraic stacks}, arXiv:1504.06467v1.


 \bibitem[AM69]{AM}
M.F. Atiyah, I.G. Macdonald:
{\it Introduction to commutative algebra,} Addison-Wesley Publishing Co., Reading, Mass.-London-Don Mills, Ont. 1969 ix+128 pp.


\bibitem[Bar03]{Bar}E.M. Baruch, \textit{A proof of Kirillov's conjecture},
Annals of Mathematics, \textbf{158}, 207-252 (2003).


\bibitem[\href{http://www.math.tau.ac.il/~bernstei/Publication_list/publication_texts/Bernstein-P-invar-SLN.pdf}{Ber84}]{BerP}
J.N. Bernstein, \emph{P-invariant distributions on GL(N) and the classification of unitary representations of GL(N) (non-Archimedean case)}, Lie group representations, II (College Park, Md., 1982/1983), 50-102,
Lecture Notes in Math., 1041, Springer, Berlin, (1984).



\bibitem[Dre00]{Dre}
J.M.Drezet, {\it Luna's Slice Theorem And Applications}, 23d
Autumn School in Algebraic Geometry "Algebraic group actions and
quotients" Wykno (Poland), (2000).

% \bibitem[Fli96]{Fli}
% Y.Z. Flicker:{\it
% Orbital integrals on symmetric spaces and spherical characters,}
% J. Algebra \textbf{184} (1996), no. 2, 705-754.
% %?? Need it? Cite it!

\bibitem[GK75]{GK} I.M. Gelfand, D. Kazhdan: {\it Representations of the group ${\rm GL}(n,K)$ where $K$ is a local field}, Lie groups and their representations (Proc. Summer School, Bolyai Janos Math. Soc., Budapest, 1971), pp. 95-118. Halsted, New York (1975).

\bibitem[HC63]{HCBul} Harish-Chandra: {\it
Invariant eigendistributions on semisimple Lie groups,}
Bull. Amer. Math. Soc. \textbf{69}, 117-123 (1963).

\bibitem[HC65]{HCReg} Harish-Chandra: {\it
Invariant eigendistributions on a semisimple Lie group},
Trans. Amer. Math. Soc. \textbf{119} 457-508 (1965).

%\bibitem[HC78]{HCQ} Harish-Chandra {\it
%Admissible invariant distributions on reductive p-adic groups., Lie Theories and their
%Applications}, Queen’s Papers in Pure and Appl. Math., Queen’s University, Kingston,
%Ontario, 281-347 (1978).
%
%\bibitem[HC81]{HC_sub} Harish-Chandra {\it
%A submersion principle and its applications}, in {\it Geometry and analysis - papers dedicated to the memory of V. K. Patodi}, Springer-Verlag, 95-102
%(1981).





\bibitem[H\"or90]{Hor} L. H\"ormander, {\it The analysis of linear partial differential operators. I. Distribution theory and Fourier analysis.}
Grundlehren der Mathematischen Wissenschaften {\bf 256}. Springer-Verlag, Berlin, 1990.



\bibitem[JR96]{JR}
H. Jacquet, S. Rallis: {\it Uniqueness of linear periods.},
Compositio Mathematica , tome \textbf{102}, no.~1 (1996), pp 65-123.


\bibitem[KV96]{KV}  J. A.C. Kolk and V.S. Varadarajan, \textit{On the transverse symbol of vectorial distributions and some
applications to harmonic analysis}, Indag. Mathem., N.S., 7 (l), pp 67-96 (1996).


\bibitem[Sak16]{Sak} Y. Sakellaridis, \emph{The Schwartz space of a smooth semi-algebraic stack}, Selecta Mathematica {\bf 22}, n. 4, 2401-2490 (2016). DOI: 10.1007/s00029-016-0285-3 (2016). See also arXiv:1505.02169.


% \bibitem[Ser64]{Ser}
% J.P. Serre: {\it Lie Algebras and Lie Groups}, Lecture Notes in Mathematics, \textbf{1500}, Springer-Verlag,
% New York (1964).

\bibitem[Sha74]{Sha} J.A. Shalika, Multiplicity one theorem for GLn, Ann. Math. \textbf{100} (1974), 171-193.

\bibitem[SZ12]{SZ}B. Sun, C.-B. Zhu: \textit{Multiplicity one theorems: the
archimedean case}, Annals of Mathematics {\bf 175}, no. 1, 23-44 (2012).





% \bibitem[vD71]{HcHvDJ}
% G. van Dijk:
% {\it Some recent results of Harish-Chandra for $p$-adic groups,} Colloque sur les Fonctions Spheriques et la Theorie des Groupes (Univ. Nancy, Nancy, 1971), Exp. No. 2, 7 pp. Inst. Elie Cartan, Univ. Nancy, Nancy, 1971.


\end{thebibliography}


\title{Invariant generalized functions supported on an orbit}
\author{Avraham Aizenbud}
%\address{Avraham Aizenbud, Faculty of Mathematics
%and Computer Science, The Weizmann Institute of Science POB 26,
%Rehovot 76100, ISRAEL.}
\address{Avraham Aizenbud,
Faculty of Mathematics and Computer Science, Weizmann
Institute of Science, 234 Herzl Street, Rehovot 7610001 Israel}
\email{aizenr@gmail.com}
\urladdr{http://aizenbud.org}
\author{Dmitry Gourevitch}
\address{Dmitry Gourevitch, The Incumbent of Dr. A. Edward Friedmann Career Development Chair in Mathematics, Faculty of Mathematics and Computer Science, Weizmann
Institute of Science, 234 Herzl Street, Rehovot 7610001 Israel}
\email{dimagur@weizmann.ac.il}
\urladdr{http://www.wisdom.weizmann.ac.il/~dimagur}

\keywords{Distribution, Nash stack, slice to a group action.\\ MS Classification: 22E45, 46F10, 14L24}
%\subjclass[2010]{20G05, 20G25, 22E35, 46F99}
%
%\classification{20G05,20G25,46F99}
%
%20-xx  Group theory and generalizations 20Cxx Representation
%theory of groups 20C99 None of the above, but in this section
%
%20Gxx Linear algebraic groups (classical groups) 20G05
%Representation theory 20G25 Linear algebraic groups over local
%fields and their integers
%
%
%22-xx  Topological groups, Lie groups 22Exx Lie groups 22E45
%Representations of Lie and linear algebraic groups over real
%fields: analytic methods {For the purely algebraic theory, see
%20G05}
%
%
%46-xx  Functional analysis 46Fxx Distributions, generalized
%functions, distribution spaces 46F10 Operations with distributions
%
%14-xx  Algebraic geometry 14Lxx Algebraic Groups 14L24 Geometric
%invariant theory [See also 13A50] 14L30 Group actions on varieties
%or schemes (quotients) [See also 13A50, 14L24]
%22E35          Analysis on $p$-adic Lie groups
\date{\today}
\maketitle

\begin{abstract}

We study the space of invariant generalized functions supported on an orbit of the action of a real algebraic group on a real algebraic manifold. This space is equipped with the Bruhat filtration. We study the generating function of the dimensions of the filtras, and give some methods to compute it. To illustrate our methods we compute those generating functions for the adjoint action of $\GL_3(\bC)$. Our main tool is the notion of generalized functions on a real algebraic stack, introduced recently in \cite{Sak}.
\end{abstract}

\tableofcontents

\section{Introduction}\label{sec:intro}
The study of invariant distributions plays an important role in representation theory and related topics (see e.g. \cite{HCBul,HCReg,GK,Sha,BerP,JR,Bar,AGRS,AG_HC,AG_MOT,SZ}). In many cases this study can be reduced to the consideration of distributions supported on a single orbit (see e.g. \cite[\S 1.5]{BerP}, \cite{KV}, \cite[Appendix D]{AG_HC}, \cite[Appendix B]{AG_ST}). While  for  non-Archimedean fields this case is very simple, for  Archimedean fields it is much more involved. In this paper we establish some infrastructure in order to analyze the Archimedean case.

Let a Nash\footnote{Nash manifolds are generalizations of real algebraic manifolds. In most places in this paper the reader can safely replace the word Nash by ``smooth real algebraic''. For more details on Nash manifolds and Schwartz functions over them see \cite{AG}.} group $G$ act on a Nash manifold $M$. Let $O$ be an  orbit of $G$ in $X$. The space $\cG(X\smallsetminus(\bar O\smallsetminus O))^G$ of tempered $G$-invariant generalized functions defined in a neighborhood of $O$ and supported in $O$ is equipped with the Bruhat filtration (see e.g. \cite{AG}).
Let $\bar\delta_{\cO}^X(i)$ denote the dimension of the $i$-th filtra and $$\bar{\mathfrak{G}}_{\cO}^X(t):=(1-t)\sum_i t^i\bar\delta_{\cO}^X(i)$$  denote the corresponding generating function.

In this paper we introduce several techniques for the computation of this function. We illustrate our methods on the case of  the adjoint action of $\GL_3(\bC)$. Our main tool is the notion of generalized functions on a real algebraic stack, introduced recently in \cite{Sak}.
\subsection{Results}
\begin{enumerate}
\item In the case of when $O$ is (locally) a fiber of a $G$-invariant submersion we prove that $\bar \fG_{O}^X(t)=(1-t)^{\dim O-\dim X}$ (see Corollary \ref{cor:reg}). \item We prove that  $\bar\delta_{\cO}^X(i)-\bar\delta_{\cO}^X(i-1)$ is bounded by  $\dim (\Sym^i(N_{\cO,x}^X))^{G_x}$ (see Lemma \ref{lem:obv}), and in the case when the stabilizer of a point in $O$ is reductive, this bound is achieved (see Theorem \ref{thm:HC})

\item  We prove that  $\bar \fG_{O}^X(t)$ is multiplicative in an appropriate sense (see Lemma \ref{lem:obv}).
\item In the general case we reduce the computation of $\bar \fG_{O}^X(t)$ to the computation of certain subspace of  distributions supported on a point in a manifold of dimension $\dim X-\dim O$ (see Theorem \ref{thm:main}). Under certain connectivity assumptions this can be reduced to an infinite dimensional linear algebra problem (see Corollary \ref{cor:main}).
\item For the case of  the adjoint action of $\GL_3(\bC)$ on its Lie algebra (or equivalently on itself) we compute $\bar \fG_{O}^X(t)$ for all orbits (see \S \ref{sec:Exm}).
\end{enumerate}

\subsection{Ideas in the proof}
Results (2,3) follow easily from the existing knowledge on invariant distributions. Result (1) follows easily from (4). Result (5) is a computation based on (4).
In order to formulate and prove Result (4) we use \cite{Sak}. Namely  we find a different presentation of the quotient stack  $G\backslash X$, and use the fact that the space of generalized functions on a stack does not depend on the presentation (See \cite[Theorem 3.3.1]{Sak}). In order to compute generalized functions in the new presentation we  replace our groupoid structure by an infinitesimal one. We do it in Theorem \ref{thm:groupoid}.

\subsection{Structure of the paper}

In \S \ref{sec:Prel} we fix notation for generalized functions on Nash manifolds, Nash groupoids and Nash stacks.

In \S \ref{sec:Group} we analyze generalized functions on groupoids. We prove Theorem \ref{thm:groupoid} which states that, under certain continuity assumptions, generalized functions on a groupoid are generalized functions on the objects manifold, satisfying a certain system of PDE.

In \S \ref{sec:GrowthFn} we define the function $\bar{\mathfrak{G}}_{\cO}^X(t)$, which is the main object of study in this paper, and establish its basic properties.

In \S \ref{sec:Nash} we introduce the stack slice, which is our main geometric tool for the computation of $\bar{\mathfrak{G}}_{\cO}^X(t)$.

In \S \ref{sec:Orb} we present a method to compute $\bar{\mathfrak{G}}_{\cO}^X(t)$ using the stack slice. We implement this method for regular orbits.

In \S \ref{sec:Exm} we compute $\bar{\mathfrak{G}}_{\cO}^X(t)$ for the adjoint action of $\GL_3(\bC)$.


\subsection{Acknowledgements}
We thank Joseph Bernstein, Bernhard Kroetz and Siddhartha Sahi for motivating questions, and Yiannis Sakellaridis for explaining us his work \cite{Sak}.

A.A.  was partially supported by ISF grant 687/13.
D.G. was partially supported by ERC StG grant 637912, and ISF grant 756/12.

\section{Preliminaries on generalized functions}\label{sec:Prel}
In this section we fix some notation %and give a short overview on
concerning generalized functions on manifolds, and tempered generalized functions on Nash manifolds and Nash stacks. We refer the reader to \cite{Hor,AG,Sak} for more details.

For a smooth manifold $M$ we denote by $C^{-\infty}(M)$ the space of generalized functions, i.e. continuous functionals on the space of compactly supported smooth measures. If $M$ has a fixed smooth invertible measure then this space can be identified with the space of distributions on $M$.

For a smooth real algebraic manifold (or, more generally, a Nash manifold) $M$ we denote by $\Sc(M)$ the space of Schwartz functions on $M$ (see e.g. \cite{AG}), and by $\Sc^*(M)$  the dual space. We call the elements of  $\Sc^*(M)$ \emph{tempered distributions} (Schwartz distributions in \cite{AG}). We also denote by $\cG(M)$ the space of tempered generalized functions, i.e. functionals on the space of Schwartz measures $\Sc(M,D_M)$ (see \cite{AG}).

For a distribution or a generalized function $\xi$ on a manifold $M$ we denote by $\WF(\xi)$ its wave-front set (see \cite[\S 8.1]{Hor}).

\begin{defn}$\,$
%\begin{enumerate}[(i)]
A Lie (resp. Nash) groupoid is a diagram $\{Mor \st{s}{t} Ob\}$ of smooth (resp. Nash) manifolds such that $s$ and $t$ are submersions,  a smooth (resp. Nash) composition map  $\mathrm{comp}: Mor\times_{Ob}Mor\to Mor$,  a smooth (resp. Nash) identity section $\mathrm{I}:Ob \to Mor$ and a smooth (resp. Nash) inversion map $inv:Mor\to Mor$ satisfying the usual groupoid axioms.
%\end{enumerate}
\end{defn}

\begin{defn}
 A generalized function $\xi\in C^{-\infty}(S)$ on a Lie groupoid $S=\{Mor \st{s}{t} Ob\}$ is a generalized function on $Ob$ such that $t^*\xi=s^*\xi$.
If $S$ is a Nash groupoid, we also define the space $\cG(S)$ of tempered generalized functions in a similar way.
\end{defn}
In \cite[Theorem 3.3.1]{Sak} it is shown that $\cG(S)$ depends only on the Nash stack corresponding to $S$ (see \cite[\S 2.2]{Sak} for the definition of the Nash stack corresponding to a Nash groupoid). Note that \cite{Sak} uses the notation $\Sc$ for Schwartz measures and $\Sc^*$ for generalized functions.

\section{Generalized functions on smooth groupoids}\label{sec:Group}
\setcounter{lemma}{0}



\begin{thm}\label{thm:groupoid}
Let $S=\{Mor \st{s}{t} Ob\}$ be a Lie groupoid. Let $\xi\in C^{-\infty}(Ob)$. Consider the following properties of $\xi$:
\begin{enumerate}
\item \label{it:xiS} $\xi \in C^{-\infty}(S).$
\item \label{it:open}For any open subset $U\subset Ob$ and any section $\varphi:U\to Mor$ of $s$ such that $\psi:=t\circ \varphi:U \to Ob$ is an open embedding we have $\psi^*\xi=\xi|_U$.
%\item \label{it:emb} For  any open subset $U\subset Ob$ and any open embedding %$\psi:U\into Ob$ such that for any $x\in U$ there is a morphism $m:x \to %\psi(x)$, we have $\psi^*\xi=\xi|_U$.

\item \label{it:temp} For any $m\in Mor,$ there exist smooth manifolds $U,V$ and a submersion $\varphi:V\times U \to Mor$ with $m\in \Im \varphi$ such that for any $x\in V$, the maps $\varphi_x^s:=s\circ\varphi|_{\{x\}\times U}$ and  $\varphi_x^t:=t\circ\varphi|_{\{x\}\times U}$ are open embeddings and we have $(\varphi_x^t)^*\xi=(\varphi_x^s)^*\xi.$

\item \label{it:VF} For any section $\alp$ of $I^*TMor$, where $I:Ob\to Mor$
is the identity section, with $ds(\alp)=0$ we have $dt(\alp)\xi=0$. Here, $dt(\alp)$ and $ds(\alp)$ are the vector fields given by $ds(\alp)_x:=d_{Id_x}s(\alp_x)$, $dt(\alp)_x:=d_{Id_x}t(\alp_x).$  \end{enumerate}
Then \eqref{it:xiS}$\Leftrightarrow$\eqref{it:open}$\Leftrightarrow$\eqref{it:temp}$\Rightarrow$\eqref{it:VF} and if for all $x\in Ob,\, s^{-1}(x)$ is connected then \eqref{it:temp}$\Leftrightarrow$\eqref{it:VF}.
\end{thm}

For the proof we will need the following lemmas.

\begin{lem}\label{lem:prod}
Let $X,Y$ be smooth manifolds. Let $\xi\in C^{-\infty}(X\times Y)$ such that for any $x\in X$, $\WF(\xi)\cap CN_{\{x\}\times Y}^{X\times Y}\subset \{x\}\times Y$ and $\xi|_{\{x\}\times Y} = 0$. Then $\xi=0$.

Here, the restriction $\xi|_{\{x\}\times Y}$ is in the sense of \cite[Corollary 8.2.7]{Hor}.
\end{lem}

This lemma follows from the next one in view of \cite[Theorem 8.2.4 and the proof of Theorem 8.2.3]{Hor}.

\begin{lem}\label{lem:prod2}
Let $V=\R^n,W=\R^k$ be real vector spaces. Let $\xi\in C^{-\infty}(V\times W)$ such that for any $x\in V$, $\WF(\xi)\cap CN_{\{x\}\times W}^{V\times W}\subset \{x\}\times W$. Fix  Lebesgue measures $V$ and $W$.
Let $f\in C_c^{\infty}(V\times W)$. Let $e_i\in  C_c^{\infty}(V\times W)$ be a sequence satisfying $\int_{V\times W} e_i(z) dz = 1$ and $e_i(z)=0$ for any $z$ with $||z||>1/i$.
For any $x\in V$ denote $g(x):=\langle \xi|_{\{x\}\times W},f|_{\{x\}\times W}\rangle$ and $g_n(x):=\langle (\xi*e_n)|_{\{x\}\times W},f|_{\{x\}\times W}\rangle$.
 Then $g_n{\to} g$ uniformly as ${n\to \infty}$.
\end{lem}
\begin{proof}
Let $U\supset \Supp f$ be an open centrally symmetric set with compact closure.
Let
\begin{multline*}\Gamma:=\left (V \times pr_{W\times (V\times W)^*}\left(\WF(\xi) \cap (\bar U\times (V\times W)^*)\right)\right)\cup \left (((V \times W) \smallsetminus U) \times (V\times W)^*\right)\\\subset  T^*(V \times W).
\end{multline*}
For any $x\in V$ denote $\xi_x:=Sh_x(\xi)$, where $Sh_x$ is the translation by $x$.
Denote $$C^{-\infty}_{\Gamma}(V\times W):=\{\eta\in C^{-\infty}(V\times W), \, \WF(\eta) \subset \Gamma\},$$
with the topology of \cite[Definition 8.2.2]{Hor}.
It is easy to see that $x \mapsto \xi_x$ defines a continuous map $V \to C^{-\infty}_{\Gamma}(V\times W)$. Let
$\xi_{n,x}:=\xi_x *e_n.$ The proof of \cite[Theorem 8.2.3]{Hor} implies that $\xi_{n,x} \to \xi_x$ as $n \to \infty$ in the topology of $C^{-\infty}_{\Gamma}(V\times W)$ uniformly in $x$. Thus, by \cite[Theorem 8.2.4]{Hor}, $\xi_{n,x}|_{\{0\}\times W}\to \xi_x|_{\{0\}\times W}$ as $n \to \infty$ in the weak topology of $C^{-\infty}(W)$ uniformly in $x$. This implies the assertion.
\end{proof}

The following lemma is standard.
\begin{lem}
Let $\varphi:M\to N$ be a submersion of smooth manifolds with connected fibers.
Let $s_0,s_1:N\to M$ be its (smooth) sections. Then, for any $y\in N$, there exists an open neighborhood $U$ of $y$ and a smooth homotopy $h:[0,1]\times U \to M$ such that $h|_{\{0\}\times U}=s_0$, $h|_{\{1\}\times U}=s_1$, and $h|_{\{t\}\times U}$ is a section of $\varphi$ for any $t$.
\end{lem}

\begin{cor}\label{cor:hom}
Let $\varphi_1:M_1\to N$ and $\varphi_2:M_2\to N$ be submersions of smooth manifolds. Assume that all the fibers of $\varphi_2$ are connected. Let $\psi_1,\psi_2:M_1\to M_2$ be smooth maps of $N$-manifolds (that is, smooth maps such that $\varphi_2\circ \psi_i=\varphi_1$).
 Then, for any $y\in M_{1}$, there exists an open neighborhood $U$ of $y$ and a smooth homotopy $h:[0,1]\times U \to M_{2}$ such that $h|_{\{0\}\times U}=\psi_0$, $h|_{\{1\}\times U}=\psi_1$, and $h|_{\{t\}\times U}$ is a map of $N$-manifolds.

\end{cor}


\begin{proof}[Proof of \Cref{thm:groupoid}]$\,$
\begin{itemize}
\item [\eqref{it:xiS}$\Rightarrow$\eqref{it:open}:]by functoriality of the pullback.
\item [\eqref{it:open}$\Rightarrow$\eqref{it:temp}:]
It is enough to show that  for any $m\in Mor,$ there exist smooth manifolds $U,V$ and a submersion $\varphi:V\times U \to Mor$ with $m\in \Im \varphi$ such that for any $x\in V$, the maps $\varphi_x^s$ and  $\varphi_x^t$ are open embeddings.
Since $s$ and $t$ are submersions, we can decompose $T_m(Mor)=V'\oplus U'$ such that $d_ms|_{U'}$ and $d_mt|_{U'}$ are isomorphisms. Let $\varphi': T_m(Mor)\to Mor$ be such that $\varphi'(0)=m$ and $d\varphi'=Id$.  By the implicit function theorem one can choose open subsets $U\subset U'$ and $V \subset V'$ such that $\varphi:=\varphi'|_{V\times U}$ is a submersion and the maps $\varphi_x^s$ and  $\varphi_x^t$ are open embeddings.


%Let $U\subset Ob$ and let $\psi:U\into Ob$ be an open embedding  such that %for any $x\in U$ there is a morphism $m:x \to \psi(x)$. For any $x\in U$ %we can find an open subset $V\subset \varphi(U)$ that includes $x$ and a %section $\lambda:V \to Mor$ of $t$ such that $s(\lambda(\psi(x)))=x$. Let %$\varphi:=\lambda \circ \psi|_{\psi^{-1}(V)}$.



\item [\eqref{it:temp} $\Rightarrow$  \eqref{it:xiS}:] by \Cref{lem:prod}.
\item [\eqref{it:open} $\Rightarrow$ \eqref{it:VF}:] Let  $\alp$ be a section of $I^*TMor$ with $ds(\alp)=0$. Define a vector field $\beta$ on $Mor$ by $$\beta_m:=(d_{I(t(m)),m}comp)(\alp_{t(m)},0),$$
where $comp:Mor\times_{Ob}Mor\to Mor$ is the composition map. By the existence and uniqueness theorem for ODE, we have an open neighborhood $\cO$ of $Mor\times \{0\}$ in $Mor \times \R$ and a map $B:\cO\to Mor$ that solves the ODE defined by $\beta$. Fix $x\in Ob$. There exists a neighborhood $U$ of $x$ and $\eps>0$ such that $U\times (-\eps,\eps) \subset \cO$.  For any $r\in (-\eps,\eps)$ define $\varphi_r(x):=B(x,r)$. Define $\psi_r:=t\circ \varphi_r$. By \eqref{it:open} we have $\psi_r^*\xi=\xi|_U$. On the other hand, it is easy to see that $$\frac{d}{dr}|_{r=0}\psi_r^*\xi=dt(\alp)\xi|_U.$$

\item [\eqref{it:VF} $\Rightarrow$ \eqref{it:temp},] for connected $s^{-1}(x)$: Let $m\in Mor$. By \Cref{cor:hom}, there exist an open neighborhood $V$ of $m$ and a smooth homotopy $h:[0,1]\times V \to Mor$ such that
$$h|_{\{0\}\times V}=I\circ s,\,\, h|_{\{1\}\times V}=I \text{ and }s(h(r,x))=s(x).$$ For any $r\in [0,1]$ and $u\in V$ consider
%$\overset{.}{\gamma}\in T_{\gamma(r)}Mor$.
$$\alp(r,v):=d_{h(r,v),inv(h(r,v))}comp(\frac{d}{dr}h(r,v),0)\in T_{I(t(h(r,v)))}Mor.$$
Extend $\alp(r,v)$ to a smooth section of $I^*TMor$ in a way that depends smoothly on $(r,v)$  such that $ds(\alp(r,v))=0$ for any $(r,v)\in [0,1]\times V $. By \eqref{it:VF} we have $dt(\alp(r,v))\xi=0$. Define a vector field $\beta(r,v)$ on $Mor$ by
\begin{equation}\label{=bet}
\beta(r,v)_n:=d_{I(t(n)),n}comp(\alp(r,v)_{t(n)},0).
\end{equation}
For any $v\in V$ we consider $\beta(\cdot,v)$ as a time-dependent vector field on $Mor$. By the existence and uniqueness theorem for ODE, we have an open neighborhood $\cO$ of $$Mor\times \{0\}\times V\cup \{h(r,v),r,v) \, \vert \, r\in [0,1], \, v\in V\}$$ in $Mor \times [0,1]\times V$ and a map $B:\cO\to Mor$ that solves the ODE defined by $\beta$.
Let $$\Xi:=I\times Id_{[0,1]\times V}^{-1}(\cO)\subset Ob \times [0,1]\times V,  A:=B\circ (I\times Id_{[0,1]\times V}|_{\Xi}) \text{ and }C:=t\circ A.$$
Let $U$ be a neighborhood of $s(m)$ in $Ob$ such that $U \times [0,1]\times V \subset \Xi$. Define $\varphi:U \times V \to Mor$ by $\varphi:=A|_{U \times \{1\}\times V }$. It is enough to prove that for any $x\in V$:
\begin{enumerate}[(i)]
\item \label{it:sOpEmb} The map $\varphi_x^s:=s\circ\varphi|_{\{x\}\times U}$ is an open embedding.
\item \label{it:tOpEmb} The map  $\varphi_x^t:=t\circ\varphi|_{\{x\}\times U}$ is an open embedding.
\item \label{it:tsxi} We have $(\varphi_x^t)^*\xi=(\varphi_x^s)^*\xi.$
\end{enumerate}
Note that $\varphi_x^s=Id$ and thus \eqref{it:sOpEmb} holds.


For $(v,r)\in V\times [0,1]$ let $\gamma(v,r):=dt(\alp(v,r))$ be a vector field on $Ob$. By \eqref{it:VF}, $\gamma(v,r)\xi=0$. It is easy to see that $$\frac{\partial}{\partial r}C(x,v,r)=\gamma(v,r)C(x,v,r).$$
Thus, for any $(x,v)\in Ob\times V$, the (partially defined) curve $C(x,v,\cdot)$ is a  solution of the ODE defined by the time-dependent vector field $\gamma(v,\cdot)$.
Note that $\varphi_x^t=C|_{U\times \{x\}\times \{1\}}$ and thus \eqref{it:tOpEmb} holds.

Finally, \eqref{it:tsxi} follows from the equality $\gamma(v,r)\xi=0$.
%
% Let $m\in Mor$. Let $\gamma:\R\to Mor$ be a smooth path such that $\gamma(0)=I(s(m))$, $\gamma(1)=m$ and $s\circ \gamma$ is constant.    Consider
% %$\overset{.}{\gamma}\in T_{\gamma(r)}Mor$.
% $$\alp(r):=d_{\gamma(r),inv(\gamma(r))}comp(\overset{.}{\gamma}(r),0)\in T_{I(t(\gamma(r)))}Mor.$$
% Extend $\alp$ to a time-dependent section of $I^*TMor$ such that $ds(\alp)=0$. Define a vector field $\beta$ on $Mor$ by $$\beta(r)_m:=d_{I(t(m)),m}comp(\alp(r)_{t(m)},0).$$
% By the existence and uniqueness theorem for ODE, we have an open neighborhood $\cO$ of $Mor\times \{0\}\cup \{(\gamma(t),t) \, \vert \, t\in \R\}$ in $Mor \times \R$ and a map $B:\cO\to Mor$ that solves the ODE defined by $\beta$.
% By \eqref{it:VF} we have $dt(\alp(r))\xi=0$. For any $m\in Mor$,
% $$d_mt(\beta(r)_m)=d_{I(t(m),m)}(\alp(r)_{t(m)}).$$
% Thus $\beta(r)(t^*\xi)=0$.
% ?? thus the evolution does not change $\xi$, but at 1 it is not an embedding. Probably we should have extended the vector field in a non-arbitrary way.
\end{itemize}
\end{proof}

% \begin{prop}
% Let $S=\{Mor \st{s}{t} Ob\}$ be a Nash groupoid and let $\fS$ be the corresponding Nash stack. Then $\cG(\fS)\cong C^{-\infty}(S)\cap \cG(Ob)$.
% \end{prop}

\section{The dimension growth function of an orbit}\label{sec:GrowthFn}
Let a Nash group $G$ act on a Nash manifold $X$.
Let $\cO\subset X$ be an orbit. Let $F_i$ be the Bruhat filtration on $\cG_{\cO}(X\smallsetminus(\bar \cO\smallsetminus \cO))$ (see \cite[Corollary 5.5.4]{AG}). Let $$V_i:=\{\xi \in F_{i} \, \vert \, \exists \eta\in \cG(X) \text{ s.t. }\eta|_{(X\smallsetminus(\bar \cO\smallsetminus \cO))}=\xi\}.$$
%
Define the distributional dimension growth function of $\cO$ in $X$ by $$\delta_{\cO}^X(i):=\dim V_i.$$
Define also the distributional normal dimension of $\cO$ in $X$ by
$$Ddim(\cO,X):=\limsup_i \frac{\ln \delta_{\cO}^X(i)}{\ln i},$$
and the distributional normal degree of $\cO$ in $X$ by
$$Ddeg(\cO,X):=\limsup_i \left(Ddim(\cO,X)! \delta_{\cO}^X(i){i^{-Ddim(\cO,X)}}\right).$$

Define  the  distributional dimension generating function by $$\mathfrak{G}_{\cO}^X(t):=(1-t)\sum_i t^i\delta_{\cO}^X(i).$$

Finally, define  the reduced versions of the above notions by
 $$\bar\delta_{\cO}^X:=\delta_{\cO}^{X\smallsetminus(\bar \cO\smallsetminus \cO)} , \, \overline{Ddim}(\cO,X):=Ddim(\cO,X\smallsetminus(\bar \cO\smallsetminus \cO)), \, \overline{Ddeg}(\cO,X):=Ddeg(\cO,X\smallsetminus(\bar \cO\smallsetminus \cO))$$

For a point $x\in \cO$ we will denote
\begin{multline*}\delta_{x}^X(i):=\delta_{\cO}^X(i), \quad \bar \delta_{x}^X(i):=\bar \delta_{\cO}^X(i), \quad  Ddim(x,X):=Ddim(\cO,X), \\ \overline{Ddim}(x,X):=\overline{Ddim}(\cO,X),\quad Ddeg(x,X):=Ddeg(\cO,X), \quad \overline{Ddeg}(x,X):=\overline{Ddeg}(\cO,X).
\end{multline*}

The following lemma follows from \cite[Corollary 5.5.4]{AG} and \cite[Corollary 2.6.3]{AG_DR}.
\begin{lemma}\label{lem:obv}
Let $x\in \cO$.  Then
\begin{enumerate}[(i)]
\item
$\delta_{\cO}^X(i)-\delta_{\cO}^X(i-1)\leq \dim (\Sym^i(N_{\cO,x}^X))^{G_x}$ and $Ddim(\cO,X)\leq\dim (N_{\cO,x}^X)$.
\item Let $U$ be an open $G$-invariant neighborhood of $\cO$ in $X$. Then $$\delta_{\cO}^U(i)\geq \delta_{\cO}^X(i) \text{ and }Ddim(\cO,U)\geq Ddim(\cO,X).$$
\item \label{it:prod} Let another Nash group $G'$ act on a Nash manifold $X'$, and $\cO'$ be an orbit. Consider the action of $G\times G'$ on $X \times X'$. Then
$$ \mathfrak{G}_{\cO\times \cO'}^{X\times X'}(t)=\mathfrak{G}_{\cO}^X(t)\mathfrak{G}_{\cO'}^{X'}(t).$$
\end{enumerate}
\end{lemma}

%\subsection{The case of reductive stabilizer}

The following theorem follows from the proof of \cite[Theorem 3.1.1]{AG_HC}.
\begin{thm}\label{thm:HC}
Let a reductive group $G$ act on an affine algebraic manifold $X$. Let $\cO\subset X$ be a closed orbit. Then $\delta_{\cO}^X(i)-\delta_{\cO}^X(i-1)=\dim (\Sym^i(N_{\cO,x}^X))^{G_x}.$
\end{thm}

\begin{remark}$\,$
\begin{enumerate}[(i)]
\item One can replace the assumption that $G$ is reductive and $X$ is affine by the weaker assumption that the stabilizer of a point $x\in \cO$ is reductive. For that one needs to use the version of the Luna slice theorem appearing in  \cite[Theorem 2.1]{StackLuna}.
\item Since the Poincare series of a finitely generated graded algebra is a rational function whose poles are roots of unity (see e.g. \cite[Theorem 11.1]{AM}), Theorem \ref{thm:HC} implies that $\mathfrak{G}_{\cO}^X(t)$ is  also a rational function whose poles are roots of unity.

\item In a similar way one can show that under the conditions of Theorem \ref{thm:HC}, the dimension  $Ddim(\cO,X)$ equals the dimension of the categorical quotient $N_{\cO,x}^X//G_x$ of $N_{\cO,x}^X$ by $G_x$.

\item Similarly, if in Lemma \ref{lem:obv} the categorical quotient $N_{\cO,x}^X//G_x$  exists then $Ddim(\cO,X)\leq\dim (N_{\cO,x}^X//G_x)$.


\end{enumerate}
\end{remark}

\section{Restriction of a Nash stack to a slice}\label{sec:Nash}
Let a Nash group $G$ act on a Nash manifold $X$.
\setcounter{lemma}{0}
\begin{defn}\label{def:StackSlice}
 Choose a point $x\in X$ and let $\cO:=Gx$ be its orbit.
\begin{enumerate}
\item We call a locally closed Nash submanifold $S\subset X$ a slice to the action of $G$ at $x$ if $x\in S$, the action map $a:G\times S \to X$ is a submersion, and $\dim \cO+\dim S=\dim X$.

\item Let $S$ be a slice to the action of $G$ at $x$. Define $M_S:=a^{-1}(S)\subset G\times S$.  Consider the quotient Nash groupoid $G\times X  {\st{pr}{a}}X$
and its subgroupoid $M_S \st{pr}{a}S$. We will call this subgroupoid a
\emph{groupoid slice} to the action of $G$ at $x$, and call
 the corresponding Nash stack a \emph{stack slice} to the action of $G$ at $x$.
%$x\stackrel[a]{b}{=}x$
%$X \st{pr}{a} X$
\end{enumerate}
\end{defn}

\begin{lem}\label{lem:ExSlice}
For any $x\in X$ there exists a slice to the action of $G$ at $x$.
\end{lem}
\begin{proof}
Choose a direct complement $W$ to $T_xX$ in $T_xGx$. It is a standard fact that there exists  a Nash manifold $S'\subset X$ containing $x$ such that $T_xS'=W$. Consider the action map $a:G \times S' \to X$. Let $S:=\{x\in S' \, \vert \,  a \text{ is a submersion at }(1,x)\}.$ It is easy to see that $S$ satisfies the conditions.
\end{proof}

The following proposition follows from the definition in \cite[\S 2.2]{Sak}.
\begin{prop}\label{prop:SliceEx}
For any $x\in X$ and any  stack slice $\fS$ to the action of $G$ at $x$ there exists an open Nash $G$-invariant neighborhood $U$ of $x$ and  such that $G\backslash U\cong\fS$.
\end{prop}



\section{Description of the space of invariant generalized functions supported on an orbit}\label{sec:Orb}

\Cref{prop:SliceEx} and \cite[Theorem 3.3.1]{Sak} imply the following theorem

\setcounter{lemma}{0}
\begin{thm}\label{thm:main}
Let a Nash group $G$ act on a Nash manifold $X$.
Let $x\in X$ such that the orbit $Gx$ is closed. Then for any groupoid slice  $\fS$ to the action of $G$ at $x$ we have a canonical isomorphism $\cG_{Gx}(X)^{G}\cong \cG_{\{x\}}(\fS)$. Here, we consider $\{x\}$ as a closed subset in $\fS$.
\end{thm}


\begin{notn}
Let a Lie group $G$ act on a smooth manifold $X$. Let $S\subset X$ be a (locally closed) smooth submanifold. Let $\varphi:S \to \fg$ be a smooth map.
For any $s\in S$ define   $\alp_{\varphi}(s)\in T_sX$ by $\alp_{\varphi}(s):=d_{e}(a_s(\varphi(s)))$, where $a_s:G \to X$ is the action map on $s$ and $e\in G$ is the unit element. Suppose that $\alp_{\varphi}$ defines a vector field on $S$, i.e. $\alp_{\varphi}(s)\in T_sS$ for any $s\in S$. Then we call this field \emph{strongly tangential to the action of $G$}.
\end{notn}

\Cref{thm:main,thm:groupoid} give the following corollary.

\begin{cor}\label{cor:main}
Let a Nash group $G$ act on a Nash manifold $X$. Let $x\in X$ such that the orbit $Gx$ is closed. Let  $S$ be a  slice  to the action of $G$ at $x$.
 Then we have a canonical embedding of $\cG_{Gx}(X)$ into the space$$\{\xi \in \cG_{\{x\}}(S)\, \vert \, \alp\xi=0  \text{ for any vector field } \alp \text{ on }S \text{ strongly tangential to the action of }G\}.$$
Moreover, if for all $x\in S,$ the set of all $g\in G$ with $gx\in S$ is connected then this embedding is an isomorphism.
\end{cor}
% - 2*a^3 + 3*a^2*la + 2*b*a - la^3 + b*la + d*e

% - la^3  + (3*a^2 +   b)*la + d*e - 2*a^3 +2*b*a

% [     a^2 + b,     2*a,     e]
% [ 2*a*b + d*e, a^2 + b,  -a*e]
% [        -a*d,       d, 4*a^2]

%\subsection{Regular case}

\begin{cor}\label{cor:reg}
Let $\varphi:X\to Y$ be a Nash submersion of Nash manifolds.
Let a Nash group $G$ act on $X$ preserving $\varphi$. Let $y\in Y$ and assume that $G$ acts transitively on the fiber $\varphi^{-1}(y)$. Then $\cG_{\varphi^{-1}(y)}(X)^{G}$ is isomorphic as a filtered vector space to $\bC[t_1,\dots,t_{\dim Y}]$.
In particular,
\begin{equation}
\fG_{\varphi^{-1}(y)}^X(t)=(1-t)^{-\dim Y}, \, Ddim(\varphi^{-1}(y),X)=\dim Y, \text{ and } Ddeg(\varphi^{-1}(y),X)=1.
\end{equation}
\end{cor}
\begin{proof}
Let $x\in \varphi^{-1}(y)$.
By \Cref{lem:ExSlice} there exists
a slice $S$ to the action of $G$ on $X$ at $x$. Shrinking $S$, we can assume that $\varphi|_S$ is an
% \et \,  map. By the inverse function theorem we can shrink $S$ further
open embedding.
Let  $M_S \st{pr}{a}S$ be as in \Cref{def:StackSlice}. By the assumption, $pr=a$. Thus the corollary follows from \Cref{thm:main}.
\end{proof}


%\subsection{Infinitesimal version}

%\subsection{Slodowy slice}

\section{Computation of $\bar \delta$ for the adjoint action  of $\GL_3(\bC)$}\label{sec:Exm}
\setcounter{lemma}{0}

\begin{thm}\label{thm:gl3}
Consider the adjoint action of $G:=\GL_3(\bC)$ on its Lie algebra $\fg$.
Let $x\in \fg$ and let $m_x$ denote its minimal polynomial. Then
$$\bar\fG_{x}^{\fg}(t)= \begin{cases}
(1-t)^{-6} & \deg m_x=3\\
(1-t)^{-6}(1+t)^{-4}(t^{2}-t+2)^2 &  m_x=(x-\lambda)^2 \\
(1-t)^{-6}(1+t)^{-2} & m_x=(x-\lambda)(x-\mu), \, \lambda\neq \mu \\
(1-t)^{-6}(1+t)^{-2}(1+t+t^2)^{-2}    & \deg m_x=1\\
\end{cases},$$
$\overline{Ddim}(\cO,X)=6$ and $\overline{Ddeg}(\cO,X)=((3-\deg m_x)!)^{-2}.$
\end{thm}

%?? Generalized functions versus distributions.

The case $\deg m_x=3$ follows from \Cref{cor:reg}. The case $\deg m_x=1$ follows from \Cref{thm:HC}. The case $m_x=(x-\lambda)(x-\mu), \, \lambda\neq \mu$ follows from \Cref{thm:HC} and \Cref{lem:obv}\eqref{it:prod}. Thus it is enough to prove the following proposition.
%?? Next version: expand!!

\begin{prop}\label{prop:gl3}
Let $G:=\GL_3(\bC)$ act on $X:=\sll_3(\bC)\smallsetminus 0$ by conjugation. Let $x\in X$ be the subregular nilpotent matrix. Then
$\bar\fG_{x}^{X}(t)=(1-t)^{-4}(1+t)^{-4}(t^{2}-t+2)^2 $.
\end{prop}

Let $e:=E_{12}\in \cO$. Let $f:=E_{21}$ and let $\fs_{\bC}:=e+\gl_3(\bC)^f$ and $\fs_{\R}:=e+\gl_3(\bR)^f$ be the Slodowy slices.

For the proof we will need the following lemma.

\begin{lem}\label{lem:SlodSlice}
$\fs_{\bC}$ is a slice for the action of $G$ at the point $e$, and for any $x\in \fs_{\bC},$ the Nash manifold $\{g\in G \, |\, gx\in \fs_{\bC}\}$  is connected.
\end{lem}
\begin{proof}
The fact that $\fs_{\bC}$ is a slice for the action of $G$ is standard. Since all the stabilizers of the action of $G$ are connected, in order to prove that $\{g\in G \, |\, gx\in \fs_{\bC}\}$  is connected it is enough to prove that the intersection of any $G$-orbit $\cO$ with $\fs_{\bC}$ is connected. For this it is enough to show that $\bar \cO\cap \fs_{\bC}$ is an irreducible algebraic variety. We divide the proof into two cases.
\begin{enumerate}[{Case} 1]
\item $\bar \cO=\{x \in X \, \vert \, \det(x-\lambda \Id)   = - \lambda^3+\gamma_1\lambda+\gamma_0  \}$ for some fixed $\gamma_0$ and $\gamma_1$. \\
Choose the following coordinates on $\fs_{\bR}$:
\begin{equation}
\fs_{\bR}=\left \{ \left(
  \begin{array}{ccc}
    a & 1 & 0 \\
    b & a & c \\
    d &0 & -2a \\
  \end{array}
\right) \right \}.
\end{equation}
%
In these coordinates, $\bar\cO \cong \{(a,b,c,d) \, \vert \, 3a^2+b= \gamma_1 \text{ and }cd - 2a^3  + 2ab=\gamma_0\}$. This variety is isomorphic to $\{(a,c,d) \, \vert \,- 8a^3 + 2\gamma_1a + cd=\gamma_0 \}.$ Thus this case follows from the irreducibility of the polynomial $- 8a^3 + 2\gamma_1a + cd-\gamma_0$ for any $\gamma_1,\gamma_2$.


\item $\bar \cO=\{x \in X \, \vert \, (x-\gamma \Id) (x+2\gamma \Id)    = 0  \}$  for some fixed $\gamma$.\\
In the coordinates above $\bar \cO$ is given by the irreducible polynomial $$cd-(2a + \gamma)^2(2a - 2\gamma).$$
\end{enumerate}
\end{proof}

\begin{lem}\label{lem:vi}
The collection of vector fields on $\fs_{\bR}$ strongly tangential to the action of $G$ is generated over $C^{\infty}(\fs_{\bR})$ by the fields $v_1,\dots,v_4$, where
\begin{align*}
v_1(A)= \left(
  \begin{array}{ccc}
    0 & 0 & 0 \\
    0 & 0 & A_{23} \\
    -A_{31} &0 & 0 \\
  \end{array}
\right), \,\, &v_2(A)= \left(
  \begin{array}{ccc}
    0 & 0 & 0 \\
    0 & 0 & -A_{11}A_{23} \\
    A_{11}A_{31} &0 & 0 \\
  \end{array}
\right)\\
v_3(A)= \left(
  \begin{array}{ccc}
    A_{31}/2 & 0 & 0 \\
    -3A_{11}A_{31} & A_{31}/2 & 9A_{11}^2-A_{21} \\
    0 &0 & -A_{31} \\
  \end{array}
\right), \,\, &v_4(A)= \left(
  \begin{array}{ccc}
    -A_{23}/2 & 0 & 0 \\
    3A_{11}A_{23} & -A_{23}/2 &0 \\
    -9A_{11}^2+A_{21}&0 & A_{23} \\
  \end{array}
\right)
\end{align*}


\end{lem}
This lemma is proven by a direct computation.
\begin{proof}[Proof of \Cref{prop:gl3}]
Let
\begin{multline*}
V_{\bC}:=\{\xi \in \cG_{\{e\}}(\fs_{\bC})\, \vert \,  \alp\xi=0 \\ \text{ for any vector field } \alp \text{ on }\fs_{\bC} \text{ strongly tangential to the action of }G\}.
\end{multline*}
and
\begin{multline*}
V_{\bR}:=\{\xi \in \cG_{\{e\}}(\fs_{\bR})\, \vert \,  \alp\xi=0 \\ \text{ for any vector field } \alp \text{ on }\fs_{\bR} \text{ strongly tangential to the action of }\GL_n(\R)\}.
\end{multline*}
By \Cref{lem:SlodSlice,cor:main} $\cG_{\cO}(X)\cong V_{\bC}$. It is easy to see that $V_{\bC}\cong V_{\bR}\otimes V_{\bR}$ as a filtered vector space. By \Cref{lem:vi},
$$V_{\bR}=\{\xi \in \cG_{\{x\}}(\fs_{\bR})\, \vert \, v_i\xi=0 \,\, \forall 1\leq i\leq 4 \}.$$
Choose the following coordinates on $\fs_{\bR}$:
$$\fs_{\bR}=\left \{ \left(
  \begin{array}{ccc}
    a & 1 & 0 \\
    b & a & c \\
    d &0 & -2a \\
  \end{array}
\right) \right \}.$$

In these coordinates we have

\begin{equation*}
v_1=c\frac{\partial}{\partial c}-d\frac{\partial}{\partial d}, \,\, v_2=-av_1,\,\, v_3=\frac{d}{2}\frac{\partial}{\partial a}-3ad\frac{\partial}{\partial b}+(9a^2-b)\frac{\partial}{\partial c}, \,\,v_4=-\frac{c}{2}\frac{\partial}{\partial a}+3ac\frac{\partial}{\partial b}+(b-9a^2)\frac{\partial}{\partial d}
\end{equation*}

Fix a Lebesgue measure on $\fs_{\R}$. It defines the generalized function $\delta_e\in \cG_{\{e\}}(\fs(\R))$. Let  $$ \delta_{ijkl}:=\left( \frac{\partial}{\partial c}\right)^i\left( \frac{\partial}{\partial c}\right)^j\left( \frac{\partial}{\partial c}\right)^k\left( \frac{\partial}{\partial c}\right)^l\delta_e.$$
If one of the indices $i,j,k,l$ is negative we set $ \delta_{ijkl}:=0$.
We have
% (?? since $x^k\frac{\partial}{\partial x}^n\delta=(-1)^kn...(n-k+1)\frac{\partial}
% {\partial x}^{n-k}\delta$)
\begin{align*}
v_1 \delta_{ijkl}&=-(k+1)\delta_{ijkl}+(l+1)\delta_{ijkl},\\
v_3\delta_{ijkl}&=-\frac{l}{2}\delta_{i+1,j,k,l-1}-3il\delta_{i-1,j+1,k,l-1}+9i(i-1)\delta_{i-2,j,k+1,l}+
j\delta_{i,j-1,k+1,l}\\
v_4\delta_{ijkl}&=\frac{k}{2}\delta_{i+1,j,k-1,l}+3ik\delta_{i-1,j+1,k-1,l}-9i(i-1)\delta_{i-2,j,k,l+1}-j\delta_{i,j-1,k,l+1}
\end{align*}
Let $\xi=\sum c_{ijkl}\delta_{ijkl}$
and note that $v_1\xi=0$ if and only if $c_{ijkl}=0\, \forall k\neq l$. Set $\delta_{ijk}:=\delta_{ijkk}$.
Let $\xi=\sum c_{ijk}\delta_{ijk}$ we get
$$v_3\xi=\sum_{i,j\geq 0, k\geq 1} \left( -\frac{k}{2}c_{i-1,j,k}-3(i+1)kc_{i+1,j-1,k}+9(i+2)(i+1)c_{i+2,j,k-1}+(j+1)c_{i,j+1,k-1} \right)\delta_{i,j,k,k-1}$$

$$v_4\xi=\sum_{i,j,k\geq 0} \left( \frac{k+1}{2}c_{i-1,j,k+1}+3(i+1)(k+1)c_{i+1,j-1,k+1}-9(i+2)(i+1)c_{i+2,j,k}-(j+1)c_{i,j+1,k} \right)\delta_{i,j,k,k+1}$$
Here, if one of the indices $i,j,k$ is negative we set $c_{i,j,k}=0$.

We obtain that $V_{\bR}$ is the collection of all finite combinations  $\sum c_{ijk}\delta_{ijk}$ that satisfy
%old computation with confution between gen fun and dist $$c_{i-1,j,k+1}\frac{k+1}{2}-3c_{i+1,j-1,k+1}(i+1)(k+1)+9c_{i+2,j,k}(i+2)(i+1)-c_{i,j+1,k}(j+1)=0$$
$$c_{i-1,j,k+1}\frac{k+1}{2}+3c_{i+1,j-1,k+1}(i+1)(k+1)-9c_{i+2,j,k}(i+2)(i+1)-c_{i,j+1,k}(j+1)=0$$
for all $i,j,k \geq 0.$

Let $F^n$ be the Bruhat filtration on $V_{\bR}$ and $G^l$ be the filtration on $F^n(V_{\bR})$ given by $$G^l(F^n(V_{\bR}))=\left\{\sum c_{ijk}\delta_{ijk} \in F^n(V_{\bR}) \, \vert \, \forall k> l \text{ we have } c_{ijk}=0\right\}.$$
It is easy to compute that $$\dim G^l(F^n(V_{\bR})) - \dim G^{l-1}(F^n(V_{\bR}))=n-2l.$$ Thus $$\dim F^{2m}(V_{\R})=  m(m+1) \text{ and }\dim F^{2m+1}(V_{\R})=  (m+1)^2.$$
Define the power series
$$f(s):=\sum_n s^{n+1}=s/(1-s) \text{ and }g(t):=\sum _n\dim F^n(V_{\R})t^n.$$
Then
$$\sum_mm(m+1)s^m=sf''(s)=2(1-s)^{-3} \text{ and }\sum s^m(m+1)^2=(sf'(s))'=(1+s)(1-s)^{-3}.$$
We get
\begin{multline*}
g(t)=\sum_m(t^{2})^m(m+1)+t\sum_m(t^{2})^m(m+1)^2= 2(1-t^{2})^{-3} +t(1+t^{2})(1-t^{2})^{-3}\\
=(t^3+t+2)(1-t^{2})^{-3}=(t^2-t+2)(1-t)^{-3}(1+t)^{-2}
\end{multline*}
Thus $$\sum _n(\dim F^n(V_{\R})-\dim F^{n-1}(V_{\R}))t^n=(t^2-t+2)(1-t)^{-2}(1+t)^{-2},$$
and hence $$\bar\fG_{x}^{X}(t)=\sum _n(\dim F^n(V_{\bC})-\dim F^{n-1}(V_{\bC}))t^n=(t^2-t+2)^2(1-t)^{-4}(1+t)^{-4}.$$
\end{proof}



\end{document}


